Author’s response to reviews

Title: Exceptional response to chemotherapy followed by concurrent radiotherapy and immunotherapy in a male with primary retroperitoneal serous adenocarcinoma: A case report and literature review.

Authors:

Young Kwang Chae (young.chae@northwestern.edu)
Naira Saleem (nyrasaleem14@gmail.com)
Yoonhwan Roh (yoonhwanroh@gmail.com)
Haris Bilal (harisbilal00@gmail.com)
Pedro Viveiros (pedrodeviveiros@gmail.com)
Bhookmika Sukhadia (bhoomika.sukhadia@gmail.com)
Xiaoqi Lin (xlin@northwestern.edu)
Muhammad Mubbashir Sheikh (Mubashirsheikh245@gmail.com)
Lee Chun Park (2000p.lcp@gmail.com)

Version: 2 Date: 14 Jan 2019

Author’s response to reviews:

Dear Editor,

We are very pleased to know our manuscript is being considered for publication. We tried to make all the additional changes. I would like to address the editor points as follows:

1. We included a statement on ethics approval and consent to participate in the “Ethics approval and consent to participate” section of the Declarations and, as in this case report there is no methods section, we added a sentence in the final line of the ‘Case Report’ section.

2. We made sure all figures were provided as separate files.
3. We verified all figure files contained only the image, as well as any associated keys/annotations.

4. We included a statement in the authors' contributions section to the effect that all authors have read and approved the manuscript.

5. We would also like to provide information on XL and MS’s contribution to the manuscript as authors. We confirm that they participated as author after reviewing the publication’s definition and adjusted the participation disclosure in our manuscript.

XL made substantial contributions to acquiring, analyzing and interpreting data and was involved in revising the manuscript critically for important intellectual content. He also gave final approval after reviewing the final version and agreed to be accountable of all aspects of the work.

MS made substantial contribution when acquiring data and was always involved in the drafting of the manuscript and in revising it to its final form, giving final approval and being accountable for all aspects of this work.

6. We also uploaded the manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours, doing the same for all relevant tables/figures.

We would like to thank again all reviewers and editors for all their contribution.

Kind regards,

Young Kwang Chae, M.D.

Assistant Professor

Department of Medicine

Feinberg School of Medicine,

Northwestern University,

Chicago, IL- 60611, USA

Phone (Cell): (312) 926-4248

Email: young.chae@northwestern.edu